Author's response to reviews

Title: Impact of diabetes on hospital admission and length of stay among a general population aged 45 year or more: a record linkage study.

Authors:

  Elizabeth J Comino (E.Comino@unsw.edu.au)
  Mark F Harris (M.F.Harris@unsw.edu.au)
  Islam Fakhrul (M.Islam@unsw.edu.au)
  Duong T Tran (D.Trans@uws.edu.au)
  Bin Jalaludin (Bin.Jalaludin@sswahs.nsw.gov.au)
  Louisa Jorm (L.Jorm@uws.edu.au)
  Jeff Flack (Jeff.Flack@sswahs.nsw.gov.au)
  Marion Haas (Marion.Haas@chere.uts.edu.au)

Version: 5  Date: 1 December 2014

Author's response to reviews:

See attached letter